Author's response to reviews

Title: Developing a measure of muscular power during a functional task for older adults

Authors:

Michelle Gray (rgray@uark.edu)
Sally Paulson (sapaul@ship.edu)

Version: 5 Date: 25 November 2014

Author's response to reviews:

The following corrections were made to the manuscript entitled: "Developing a measure of muscular power during a functional task for older adults".

1- We have added the name of the ethics committee that approved the study.
2- We have double checked and ensured that the manuscript is formatted to the specifics of BMC.
3- The additional files have been added to the main manuscript document.